Author’s response to reviews

Title: Dynamic Predictive Accuracy of Electrocardiographic Biomarkers of Sudden Cardiac Death within a Survival Framework: The Atherosclerosis Risk in Communities (ARIC) study

Authors:

Erick Perez-Alday (Erick.PerezAlday@gmail.com)
Aron Bender (bendear@ohsu.edu)
David German (german@ohsu.edu)
Srini Mukundan (Srini_v_mukundan@rush.edu)
Christopher Hamilton (christopher.hamilton@my.rfums.org)
Jason Thomas (thomasjt@uw.edu)
Yin Li-Pershing (annabel2@pdx.edu)
Larisa Tereshchenko (tereshch@ohsu.edu)

Version: 2 Date: 18 Oct 2019

Author’s response to reviews:

Dear Dr. Seidlerová,

Thank you very much for helpful suggestions. We have revised the manuscript according to your suggestions, and followed all formatting requirements.

Sincerely,

Larisa Tereshchenko, MD, PhD.